Author's response to reviews

Title: The reduction of disability in community-dwelling frail older people: design of a two-arm cluster randomized controlled trial

Authors:

Silke F Metzelthin (s.metzelthin@zw.unimaas.nl)
Erik van Rossum (e.van.rossum@hszuyd.nl)
Luc P de Witte (l.de.witte@hszuyd.nl)
Marike RC Hendriks (marike.hendriks@beoz.unimaas.nl)
Gertrudis IJM Kempen (g.kempen@zw.unimaas.nl)

Version: 2 Date: 18 August 2010

Author's response to reviews: see over
Maastricht, 18 August 2010

Dear Dr. Norton,

We are very glad that you have accepted our manuscript for publication in BMC Public Health.

We have made the following final changes:

1. Authors' contributions: We have changed full names into initials.
2. References: We have changed reference 29 in the text in the following way: Authors: Title of manuscript, submitted.
3. We have removed table placement text.
4. Abbreviations: We have formatted the abbreviations into a paragraph.
5. Figures:
   a. We have cropped the figures to minimise the white space around the image.
   b. We have removed the title and figure number in the image files The title is part of the manuscript file after the reference list.
   c. Figure 3: I think figure 3 is already in the correct orientation (landscape).
6. Typography: Small typographical corrections and revisions were made.

We hope that all the changes we have made fulfill your expectations.

On behalf of all authors,

Yours sincerely,

Silke Metzelthin, MPhil.